Author’s response to reviews

Title: An evaluation of the integration of non-traditional learning tools into a community based breast and cervical cancer education program: the witness project of buffalo

Authors:

Dr Thelma C. Hurd MD (thelma.hurd@roswellpark.org)
Prof Paola Muti MD (muti@buffalo.edu)
Deborah O. Erwin PhD (erwindeboraho@uams.edu)
Sharita Womack PhD (sdwomack@buffalo.edu)

Version: 4 Date: 16 May 2003

Dear Sir:
Thank you for accepting our manuscript. The manuscript entitled An evaluation of non-traditional learning tools into a community based breast and cervical cancer education program: the witness project of buffalo, has been revised and the following changes made:

1. All unnecessary capitalization has been removed from the title.
2. Authors' qualifications have been removed.
3. Country for each affiliation has been added.
4. The manuscript sections have been reordered.
5. Competing interests have been declared.
6. Underlined text has been replaced with italic text.
7. References have been reformatted to fit journal criteria.

Again, thank you for review and acceptance of our manuscript.
Sincerely.
Thelma C. Hurd, MD